Author's response to reviews

Title: Ardispusilloside I induces apoptosis via regulating Bcl-2 family proteins in human mucoepidermoid carcinoma Mc3 cells

Authors:

Xiaofang Xu (dentistxxf@hotmail.com)
Taoli Zhang (taolizhang@fmmu.edu.cn)
Wang Rong (kgyyjk@fmmu.edu.cn)
Song Jin (jinsong@yahoo.com.cn)
Xin Xiao (xiaoxin@fmmu.edu.cn)
Weidong Zhang (wdzhang@fmmu.edu.cn)
Pengyuan Wang (pywang@fmmu.edu.cn)
Xiaojuan Wang (wxjyh231@fmmu.edu.cn)

Version: 5 Date: 11 April 2013

Author's response to reviews: see over
Dear editor James Prozenko and Prof Xiaolin Zi:

Thanks a lot for your time and attention to our manuscript (MS: 1364305419733130). And I'm most appreciative of the reviewers’ and your valuable comments and suggestions.

In the 4rd revision, we provided the results of both 24 and 48 hr. The revised manuscript has been edited by a native-English speaker with scientific expertise from Edanz, as your kindly recommendation. And we hope that our efforts would make the manuscript more completeness and suitable for publication. The following is our point-by-point responses to the reviewers’ comments for our manuscript.

The authors should.

Thanks again for your time and attention

Sincerely yours,

Xiao-Fang Xu
RESPONSE TO THE COMMENTS.

1. To reviewer Bungorn Sripanidkulchai

Reviewer's report:

1) The authors should provide the results of both 24 and 48 hr. To make clear the findings.

Response: Thanks for your suggestions. We have provided the results of both 24 and 48 hr for the MTT assay and the apoptosis assay tested by Hoechst 33342 staining, FCM and TEM. Also, the revised manuscript has been carefully edited by a native-English speaker with scientific expertise from Edanz.